



# Article To Align Technological Advancement and Ethical Conduct: An Analysis of the Relationship between Digital Technologies and Sustainable Decision-Making Processes

Teresa Riso and Carla Morrone \*

Department of Law and Economics of Production Activities, Sapienza University of Rome, 00161 Rome, Italy \* Correspondence: carla.morrone@uniroma1.it

Abstract: Digitalization and sustainability is widely investigated; however, only few studies have analyzed the role of sustainable decision-making processes in the business strategy field for achieving sustainable development goals: The study proposes a structured literature review (SLR), analyzing 318 documents published in the period 2019–2023. The results of the SLR bring to light that the publications on the topic mainly regard seven research areas. Green supply chains, logistics and digital management is the principal one, followed by sustainable goals, green indicators, and digital advancement. Furthermore, the analysis marks future research lines: although this document offers an overview of the main studies in literature, the major limitation is the use of only one database and a time span of 5 years. This study could contribute to generate further research on sustainable decision-making processes, promoting a different organizational approach to value creation and sustainable performance.

Keywords: sustainability; decision-making processes; digitalization; ethical conduct; digital technologies



Citation: Riso, T.; Morrone, C. To Align Technological Advancement and Ethical Conduct: An Analysis of the Relationship between Digital Technologies and Sustainable Decision-Making Processes. *Sustainability* 2023, *15*, 1911. https://doi.org/10.3390/su15031911

Academic Editors: Raffaele Fiorentino, Rosita Capurro, Barbara Iannone and Federica Ricci

Received: 23 December 2022 Revised: 13 January 2023 Accepted: 14 January 2023 Published: 19 January 2023



**Copyright:** © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

# 1. Introduction

Digitization, "as process of application of digital technologies within a firm" [1], has led organizations towards important changes in business administration by attracting great attention in the literature to the newly offered opportunities relating to the productivity, efficiency, and sustainability of processes. Firms are the main sustainable process creators, being able to internally modify the organization, the business models and the strategic choice considering economic and ethical aspects. According to Fiorentino et al. [2] and Merendino et al. [3], the essentiality of smart technologies is demonstrated; however, they are not always adopted in strategic decision-making processes due to the lack of capabilities or awareness of the opportunity that they offer.

Digital technologies could take on an enabling role in the circular economy (CE) [4], arriving "to improve the decision-making processes across different stages of the industry life cycle" [5]. In the CE, stimulating economic growth in line with environmental sustainability [6] is a crucial aspect of the application of reuse, recycling, recovery, and re-manufacturing practices [7], necessary to contribute to the creation of the sustainable development goals of Agenda 2030. In this context, digitalization should be read as a process economically, environmentally, and ethically sustainable, suited to foster firms in addressing the right direction for world wellness. Going beyond themselves is the main goal of firms that intend to fully embrace the creation of a sustainable organization. How? Firms should adapt and effectively build their capabilities to the requirements of environmental sustainability, and to do so they have a great opportunity: to use digital technologies like the internet of things, cloud computing, artificial intelligence, big data analytics [8], blockchain technology [9], next-generation telecommunication networks [10], and 3D printing. Although the role of technologies in modifying the means of decision-making is known [11], it is necessary to deepen how it takes place and how the firms can

modify the organizational elements impeding the implementation of sustainable principles. Due to the influence of digital technologies on organizational capabilities, by making them competitive and dynamic [12], the convergence of sustainability and digital technologies is beginning to gain more attention in the private and public sectors [13]. However, there is still a lack of systematic and rigorous research in this field, which should highlight how decision-makers are seeking new bases and drivers for economic growth, considering sustainable development. In fact, despite the existing literature, which tries to demonstrate how digital technologies sustain the achievement of sustainability in all its aspects, there are few studies focused on decision processes [14]. In this regard, although some recent articles try to investigate the link between sustainability and digitalization, these do not embrace and fully explain the effects on decision-makers in the context of the circular economy. Considering this, this research tries to identify new directions in the literature on the relationship between sustainable decision-making and digital technologies by using a structured literature review (SLR) [15–17]. SLR aims to establish the foundations of research in the field and reduce subjectivity, bias, and errors [18–21]. Referring to the main existing contributions, it also provides and shows some insights for future research. Through the Scopus database, 318 articles published between 2019 and 2023 were selected to investigate how digitization sustains and influences decision-making processes, providing elements that show how digitalization is functional for sustainability and how it can influence the still hidden aspects of decision-making processes in the era of circular economy.

To reach the target, the authors set the following research questions:

RQ1. What are the main research lines in which sustainability and digital technologies are investigated?

RQ2. What are the research focus and settings of interest of these research lines?

RQ3. What are the possible future trends of research?

As stated by De Felice and Petrillo [22], to ensure convergence between technological and sustainable development it is necessary to define decision-making priorities committed to sustainability. Noting that sustainability means building development able to satisfy the needs of the current generation, preserving, at the same time, the availability of resources for future ones [23], digital technologies represent a great support for social business due to their capabilities to leverage practices towards financial and operational sustainability by generating sustainable value [24].

This paper offers several implications for scholars and practitioners, revealing a link between digital technologies and sustainability through decision-making processes. Furthermore, from a theoretical perspective, this research contributes both to the emerging literature on CE and strategic management literature on the decision-makers' role in resource management. The support of digitalization in sustainable decision-making processes could lead to a different organizational approach to value creation and firm performance.

The article is structured as follows: Section 2 provides the research background for the SLR; Section 3 describes the methodology used; Section 4 presents the findings; Section 5 discusses the results and the theoretical and managerial implications; Section 6 shows the conclusion, future research directions, and limitations of the study.

#### 2. Theoretical Framework

#### 2.1. The Role of Digitization in the Circular Economy Era

The emergence of digitization. with its high potential, is the topic of several existing studies in different fields of research [25–31], due to its countless benefits to firms' behavior, changes in production methods, and organizational processes [32]. According to Nambisan et al. [33], a digital orientation promotes the acquisition of new knowledge and new competencies as well as the generation of new processes or products and fosters sustainability. Nowadays, after the COVID-19 emergency, firms are more often called to follow market requirements about CE, whose practices and strategies can lead toward increasing sustainability and resilience [34,35].

CE is defined as "a sustainable development initiative with the objective of reducing the societal production-consumption systems' linear material and energy throughput flows by applying materials cycles, renewable and cascade-type energy flows to the linear system" [36]. It aims to create "a resource-effective and resource-efficient economic system" without generating environmental destruction [37,38]; that is, it combines economic development without worsening the environmental load with economic activities [4], through the actions of reducing, reusing, recycling, recovering, and remanufacturing [39]. The CE transition has several barriers that must be overcome. Digital technology implementation can help to pursue this purpose of saving and promoting sustainability [40], since it represents an excellent ally [41,42], capable of supporting, equipping, and generating a strategy based on resource efficiency for smart firm creation [43]. According to Massaro et al. [44], digital technologies can help in achieving higher sustainability where the customers become an integral part of the process by offering their contribution [45]. Industry 4.0 [46] refers to "new technological infrastructure, interconnected set of cyber-physical systems, and advanced data management solutions enabling new manufacturing processes, value chain organizations, and strategy development paths" [47]. It is based on advanced digital technological innovations and on design principles that make it possible to define the actions required for the Industry 4.0 transition [48–50]. Digital technologies can give life to new circular business models capable of being profitable and are contributing to the dematerialization processes of products. Although the integration and the impact of CE principles in innovative business models is the focus of many studies [38], so far there is a limited amount of research that investigates the crucial elements for the success of CE in businesses [51] and why digital technologies promote the implementation of CE strategies [52]. In this regard, it is necessary to observe digital technologies' role, which is crucial in virtue of the opportunities that digital tools offer in tracking the flow of products in resource management and in decision-making processes. The relationship between new technologies and CE [53] is possible, among other things, thanks to the remarkable amount of data that they produce [4]. In recent years, a strong digitization and automation of processes took place, leading firms towards increasing awareness of digital technologies' relevance and their adoption for the creation of smarter and better-connected business models [54]. According to Akbari and Hopkins [28], if previous studies are focused on barriers to adoption and challenges, to date there is a need to explore the relationship between "current adoption" and "future impact" in order to become fully aware of the opportunities embedded in technological change, especially if the goal is becoming a sustainable organization. "Digitalization requires specific organizational structures, such as separate business units and agile organizational arrangements" [55], and sustainability requires new processes or new configurations for existing capabilities [56]. Indeed, circular business models can promote environmental improvements [57]. Regarding this, it is useful to take into consideration the four different scenarios summarized by the European Agency for Safety and Health at Work [25] and correlated implications, as reported in Table 1.

Finally, as noted by EU-OSHA [25] as well as Junge and Straube [58], the transition to a circular economy requires digital technologies that "optimize the logistics resources and energy efficiency to pave the path for CE". CE brings to attention the role of firms in the transition towards a sustainable model by acting in a proactive way on stakeholder networks [59].

#### 2.2. The Impact of Digitization on Sustainability (for Better or for Worse)

Del Vecchio et al. [60] state, "being smart also means being sustainable". Indeed, the role of sustainability is not limited to preserving the environment, but represents "a key priority aimed to drive long-term success and value creation" [61] through the development of new business models that are able—thanks to the exploitation of new technologies—to implement innovative strategies [62,63] and effective value propositions [64].

The Roaring 40 s—FullyCarbon Neutrality—Of aCircular and InclusiveHazardous Kind		Staying Afloat—Amid Economic and Environmental Crises	Regional Circularities—With European Divides					
In 2040, the products that sell best are those that are cradle-to-cradle and 'net-positive' in terms of social and environmental sustainability. Re-use takes precedent over replacement; environmental and safety considerations dominate decision-making.	The year 2040 marks the achievement of carbon neutrality in Europe. However, with environmental outcomes having been prioritized above all else, this has often come at the expense of job quality and working conditions, with workers widely dispersed and frequently alone.	In 2040, the biggest concern for many is just having a job—not what the job entails. Most people are focused on keeping things afloat, so that there's little consideration for much else—not the environment, social rights or job quality.	In 2040, everyone knows that contracted employees are well looked after, but those in non-standard employment are not. Neither is the environment, with circularity being mostly regional.					
Potential Implications for OSH in 2040 Resulting from Digitalisation in the CE								
Physical hazards are reduced across all industries, but psychosocial risks increase due to the increased digitalisation and automation in the CE. Databases for all materials and products reduce hazards in repair, reuse, and recycling.	Decentralized deployment of workers makes OSH supervision and monitoring much more difficult. Mobile work means that workers are more likely to work in unsafe environments. Rapid introduction of new materials means that documentation is insufficient, resulting in worker exposure to unknown hazards.	Workers lack the skills necessary to navigate or find employment, increasing the likelihood of employment with low OSH standards. A preponderance of platform work means that responsibility for OSH is unclear, mental health risks increase, and there is insufficient protection for freelance and contingent workers.	Workers displaced by new technologies are pushed into the informal economy with very low OSH standards. Regionally, digitalisation is very uneven, making the exchange of OSH-relevant information difficult.					

Table 1. An overview of the four scenarios and the potential OSH implications resulting from digitalisation.

Source: EU-OSHA, 2021.

Sustainable business models can be defined as business models that pay attention and direct their view towards sustainable value creation, pro-active stakeholder management, and a long-term perspective [65]. Rapid and rising attention to firms and several actors in the market regarding sustainability and, precisely, to the relationship between digitalization and sustainability, requires understanding of how organizations can reinforce a change in responsible thinking between actors when they have to carry out decision-making processes. Considering the value relevance of new technologies in all their forms, the firms anticipate getting positive sustainable and economic performance by virtue of the relationship between digital technologies and sustainability.

According to Lutfi et al. [66], the use of a digital accounting system could be an effective support to achieve excellent comprehension of the "tools adopted by firms to develop digital technology for sustainable decisions and competitive performance".

Nowadays, the increasingly strong relationship between digital technologies and sustainability have inspired the inception of "Digital sustainability" [67,68]. Digital sustainability is a concept that refers to the innovative approach adopted by firms to pursue sustainable development through digital toolboxes. In particular, George et al. [67] underline that digital sustainability supports "ecosystem-level coordination among disparate players, enabling them to work together toward shared objectives related to sustainable development". Therefore, the use of digital technologies can help to achieve the purpose of CE, "incorporating sustainability into production- and consumption-related processes" [69] and promoting value creation based on the minimization of negative environmental consequences. In agreement with Lobschat et al. [70], when firms decide to act responsibly without neglecting ethical concerns, beyond considering the multiple benefits of digitization, it is necessary to also consider the risks that the use of smart devices provoke for consumers and firms. For this purpose, at the present, further studies are needed. Bag et al. [35] try to list the main barriers that manufacturing firms run into in digital technology implementation and business sustainability, identifying technological and financial challenges as the most significant challenges, followed by process challenges, human resource challenges and collaboration challenges, security challenges, and leadership challenges.

#### 2.3. Digitization and Sustainable Decision-Making Processes: What Is the Link?

A comprehensive change in the whole organization in terms of norms, culture, and attitude is necessary to govern the development of digital technologies, which should not be considered as a single technological challenge but as an important process that involves all human resources in the organization. If the firms intend to open themselves to ethical responsibility, they should be aware of the potential and risks of digital technology use [70]. Specifically, decision-makers take on an important role in the improvement of sustainable actions thanks to the use of intelligent management systems [71]. In fact, it is the adoption of these technologies that can provide the right tools to make the firm competitive in terms of innovation and sustainability. In other words, digital contribution affects the business environment by promoting change in the decision-making procedures for new value propositions [72]. Urbinati et al. [62] underline the role of big data in the capture and value creation process by mapping the decision-making domain, while Lobschat et al. [70] address the digital corporate sustainability issue through the identification of four different stages of the lifecycle of digital technologies and data, identifying key sources of ethical responsibility, among which decision-making processes are included. They are identified as a multilevel phenomenon that is not limited to choice, but includes intelligence and design [73]. Decision-making processes allow transforming a firm's identity to make it assume a sustainable identity. In agreement with Müller and Voigt [74], digitization and data-driven decision-making processes have a positive effect on profitability, since they make it possible to exploit more information for strategic choices as well as increase the ability to make long-term decisions. This decision-making perspective is fundamental to deeply understand the influence of digital technologies on firms' behavior, so that it assumes a socio-environmental configuration. Saura et al. [40] confirm how new digital technologies can boost sustainability and circular entrepreneurship, since the development of new technology-based systems entails better data-centric decision-making processes aimed to promote sustainable actions.

#### 3. Research Methodology

This study applies a structured literature review (SLR). Through this methodology, it is possible to identify current research trends and future directions [16,17]. Thanks to the extensive availability of academic papers, the SLR made significant progress, allowing to avoid missing seminal articles and to eliminate most researcher bias [20]. The aim of the review is to analyze and to systematize all the collected articles in a replicable way. The choice of SLR is due to its versatility and the following advantages: (1) a higher quality of review process and outcomes [75]; (2) minimizing bias and errors [18,76]; (3) confirmation of results' validity due to the replication of steps during the review process [77]; (4) mapping, synthesis and organization of specific research areas [77,78]; and (5) the possibility to develop a framework about existing knowledge [18,79].

To allow the replication of the research, the SLR should follow some specific steps: (1) defining research questions (RQ); (2) writing the research protocol; (3) building the research sample; (4) a coding framework as well as the type of analysis to perform; (5) discussing and identifying future research directions [16,17,20,21,80].

The three research questions driving this study are the following:

RQ1. What are the main research lines in which sustainability and digital technologies are investigated?

RQ2. What are the research focus and settings of interest of these research lines?

RQ3. What are the possible future trends of research?

The first research question aims to identify the main existing contributions on the relationship between digital technologies and sustainability. The second research question aims to determine the state of the art. The third research question aims to pave the way for future research ideas in this field.

As mentioned, after having outlined the RQ, defining the research protocol is needed. In this regard, the research provides for the extraction of the articles through the use of the Scopus database, which is widely used and is the largest available database in the multidisciplinary scientific literature [81].

To collect all publications about sustainability and digitalization during the period from 2019 to 2023 (Figure 1), two searches were launched and then integrated:

- Search group 1: "sustainability", "digitalization" and "decision-making", obtaining a first sample of 984 items. In particular, the Scopus string used to extract the contributions is: (TITLE-ABS-KEY ("sustainability") AND ALL ("digitalization") AND ALL ("decision-making").
- 2. Search group 2: "sustainability", "digitization" and "decision-making", obtaining a second sample of 531 items. In particular, the Scopus string used to extract the contributions is: (TITLE-ABS-KEY ("sustainability") AND ALL ("digitization") AND ALL ("decision-making").



Total papers screened and eligible: 455 First research group: 293 Second research group: 162

> After eliminating not relevant articles and duplicates Total 318

Figure 1. Summary of the extraction of papers from the Scopus database.

Once all 1.515 documents available on Scopus for both groups were identified, the research strategy focused on the inclusion and exclusion criteria pertaining to the subject area, the year of publication, and the type of scientific paper.

Specifically, the study field was limited to the "Business, Management and Accounting" and "Economics, Econometric and Finance" areas. With these filters, the total number of results was reduced respectively to 376 for the first group, of which 338 were from the "Business Management and Accounting" and 90 from the "Economics, Econometric and Finance" research areas, and 207 for the second group, of which 193 were from the "Business Management and Accounting" and 39 from the "Economics, Econometric and Finance" research areas.

Afterward, the filter related to publication year was added as detailed before (2019–2023). This choice made it possible to identify recently debated topics and the literature discussed in recent years, focusing on the timespan in which the scientific debate on these topics has become more significant.

Finally, the search was narrowed to the following types: "articles", "books", and "book chapters". This choice guarantees the inclusion of documents validated by the scientific community and the disqualification of non-validated documents.

After removing duplicated and non-relevant contributions, the final number of studies for the SLR analysis was 318.

Figure 1 shows and describes the process used to identify the eligible research.

This review is based on the framework proposed by Paoloni and Demartini [82], which is articulated through three main dimensions: article focus, research method, and geographical area.

In the following section, the chosen analytical framework is described.

To classify and analyze the items, the authors used a manual approach, reading the full papers and summarizing the results. Specifically, each author read the full papers individually, sharing the results with the entire research group. This step allowed for an in-depth sharing of the content. Moreover, the manual approach allowed the authors to shed light on research trends for each group identified.

## Analytical Framework

A manual approach [83] to performing the quality assessment process was applied. Following the aforementioned framework [82], the review of the selected articles is based on three main dimensions:

- A. Article focus
- B. Research method
- C. Geographical area

Based on the article focus (A), seven groups can be identified, as also reported in Table 2:

Table 2. Shows the "focus" of eligible documents per year of publication.

Year	A1. Sustainable Goals, Green Indicators, and Digital Advancement	A2. Sustainable Business Model and Digitalization	A3. Green Supply Chain, Logistics and Digital Management	A4. Sustainable Financial and Economic Performance, Value Creation, and Digitization	A5. Sustainable Entrepreneur- ship and Digital Technologies	A6. Circular Economy and Digitization	A7. Others	Total by Year
2019	7		8					15
2020	12	4	14	4		1	2	37
2021	26	8	35	7	2	10	2	90
2022	56	9	63	25	5	10	5	173
2023			2	1				3
Total by focus	101	21	122	37	7	21	9	318

A1. Sustainable goals, green indicators, and digital advancement

This group includes articles addressing sustainability issues and sustainable development goals and explaining how digital advancement contributes to sustainable organization creation.

A2. Sustainable business model and digitalization

This group includes studies analyzing how digitalization affects the adoption of a new sustainability-oriented business model.

A3. Green supply chain, logistics and digital management

This group concerns studies that consider supply chain digital management to give it a sustainable identity. In addition, in this category, the authors included research related to the analysis of the relationship between digitalization and green logistics.

A4. Sustainable financial and economic performance, value creation, and digitization

This group includes studies examining the impact of digitalization on financial and economic performance by considering sustainability as a variable for value creation and high performance.

A5. Sustainable entrepreneurship and digital technologies

This group includes research investigating the relationship between green entrepreneurship and digital technologies and how digital tools efficiently support sustainable entrepreneurship.

A6. Circular economy and digitization

This group includes articles addressing the relationship between circular economy and digitalization. Specifically, this group includes articles investigating digital barriers and drivers in the circular economy.

A7. Others

This is a residual group that includes results of the SLR that belong to none of the previous categories. This class concerns studies that are not related to the aim of the present paper but were among the findings only because the keywords "decision-making" and "digitization" or "digitalization" were present in Scopus research fields. Therefore, these were considered outside of the focus of this paper.

Based on the research method (B), six clusters were identified:

B1. Literature analysis (e.g., structured literature review, systematic literature review);

B2. Qualitative research (e.g., case studies, content analysis, surveys through interviews); B3. Quantitative analysis (e.g., generalized method of moments (GMM); partial least-

square technique (PLS); structural equation modelling; regression modelling);

B4. Mixed research (studies with more than one method applied);

B5. Theoretical analysis (e.g., conceptual or discursive contributions);

B6. Other (research written with methods not described above).

Based on the geographical area (C), 11 categories were identified:

C1. Eastern Europe

C2. Middle East

C3. South and Central America

C4. North America

C5. Northern Europe

C6. Southern Europe

C7. Asia

C8. Africa

C9. UK

C10. Oceania

C11. Mixed

## 4. Findings

4.1. Article Focus

The authors carried out the quality assessment process [16,82,83] and progressively categorized the topics into the mentioned seven groups.

The main focuses of scholars are A1., with 101 out of 318 documents (32%), A3., with 122 documents (38%), and A4., with 37 documents (12%).

Regarding the time of publication, most research is from 2020–2022 (obviously 2023 is not cited, because publications are limited, considering that the year is yet to come.).

## 9 of 21

#### 4.2. Research Methods

The most-used research methods are B4., with 88 of 318 documents, followed by B3., with 75, B2., with 67, B1., with 47, B6., with 22 and B5., with 19. Table 3 shows all the results.

Table 3. Research Methods.

Year	B1. Literature Analysis	B2. Qualitative Research	B3. Quantitative Research	B4. Research Mix	B5. Theoretical Analysis	B6. Other	Total by Year
2019 2020 2021 2022	2 2 11 32	2 12 15 37	4 5 25 39	3 11 28 46	2 6 11	4 5 5 8	15 37 90 173
2023 Total by research method	47	1 67	2 75	88	19	22	3 318

In addition, mixed research methodology was preferred (for the whole timespan) for A3., with 39 items out of 122 documents; A1., with 27 out of 101; A4., with 9 out of 37 documents; A6., with 6 out of 21; A2., with 4 out of 21; A7., with 2 out of 9; and A5., with one result.

## 4.3. Geographical Area of Authors Affiliation

Most of the analyzed documents were written by authors with affiliations from different geographical areas (135 of 318), followed by research published by Asian scholars (59 of 318) and by authors with affiliations from Northern Europe (42 of 318), as detailed in Table 4.

Table 4. C. Geographical area.

Year	C1. East Europe	C2. Middle East	C3. South and Central America	C4. North America	C5. Northern Europe	C6. Southern Europe	C7. Asia	C8. Africa	C9. UK	C10. Oceania	C11. Mixed	Total by Year
2019	1		1		4	1	3				5	15
2020	2		1		5	4	10	1	2		12	37
2021	1	1	4	2	9	9	14	1	1	2	46	90
2022	2	5	2	9	24	18	32		8	3	70	173
2023	1										2	3
Total by geographical area	7	6	8	11	42	32	59	2	11	5	135	318

## 4.4. Documents Citations

To understand the scientific impact of the different research topics, the most-cited publications were considered, focusing on documents with a citation index (h-index) higher than 50. This resulted in 19 documents with a h-index between 50 and 294.

The average highest citation index was recorded in 2020. These most-cited studies mainly used quantitative methodologies (B3.) and mixed-method approaches (B11.). Among the most-cited documents of 2020, no research fell into the literature analysis category (B1.) even though the most quoted of all papers is a literature analysis, published in 2019. Finally, regarding the geographical area, the most-cited research was written by authors located in different geographical areas (C11.; 58%), followed by authors with affiliations from Asia (C7.; 16%) and Southern Europe (C6.; 16%).

The main research lines are described in the following section, answering RQ2.

#### 5. Discussion

This section intends to answer RQ2 (What are the research focus and settings of interest of these research lines?). The SLR quality assessment process pointed out that sustainability and digitization involve six main research areas.

Planning possible scenarios makes it possible to define the necessary behaviors to reduce risks in decision making and increase the probability of making the right decision regarding continuous changes [84].

Regarding the first group, "sustainable goals, green indicators, and digital advancement", the analysis of the results illustrates how sustainable system development represents an important aim when the needs and impacts of current and future generations are evaluated by decision-making models [85], aligning management's actions to a firm purpose. Recurring calls for a more sustainable world are increasing the need to investigate the economic, financial, and social implications of digital technologies used to solve sustainability tensions, which should not be neglected but considered "as a useful tool to sustain their business partnerships and enhance their organizational ambidexterity and, consequently, achieve holistic sustainability" [86]. Holistic sustainability as result of the utilization of sustainability knowledge is able to create the conditions for the development of sustainable innovation [87]. To analyze sustainable development goals, some studies examine the relationship between blockchain technologies and environmental sustainability [88–91], investigating the role of blockchain, which requires in-depth analysis of potential positive and negative effects on the environment. If the negative effects are explored in a few studies [2], just as key components that foster the transition from traditional systems to new sustainable systems, "blockchain for good" [92,93] is a responsible event influencing the way to operate some firms. It has the capability to support the development of renewable energy, and it is considered a way to realize sustainable behaviors in compliance with the standard established by the UN for SDGs, but not only this. Through its influence on supply chains as well as its capability to distinguish green suppliers from those who work without carrying out responsible behavior [94], it is able to affect the livability of the cities that take the form of smart cities. Smart cities promote sustainable development; for this reason, the concept of "smart sustainable cities" is mentioned many times [95]. These are the results of political work exerted by politicians, local governments, and policymakers to carry out SDGs, in particular, SDG 11: Sustainable Cities and Communities [96]. Guandalini [97] addresses the analysis of the relationship between sustainability and digitalization, finding five important gaps in the existing literature: the lack of a management focus on this relationship, misalignment of the terminology used, overarching strategic studies, scarcity of comparative research, and studies at an organizational level. In fact, the misalignment of the terminology is a relevant problem because it makes it difficult to identify clear strategic guidelines used by the firms to promote the effective implementation of digital technologies aimed at the compliance of standards established by United Nations. Industry 4.0 takes on a crucial role in promoting the transition and the achievement of sustainability in manufacturing [98–100]. However, there is still a research gap concerning Industry 4.0's impediments to sustainable digital manufacturing. Industry 4.0 has the potential to address some of the most pressing challenges for the achievement of the SDGs, so investigating Industry 4.0's main impediments to the advancement of digital manufacturing for long-term sustainability [101] is necessary to pursue and reduce the distance between Industry 4.0 and sustainable manufacturing.

Regarding the second group, "sustainable business model and digitalization", many studies focus on sustainable business model innovation, which represents the evolution of a traditional business model, which does not limit attention to resource coordination to create and deliver value to consumers and other stakeholders [102], but can sustain the implementation of ethical values into business strategy for firms' creation output [103]. Many studies examine business model innovation, and some of them [104] investigate platform business models contributing to the concept of value-based health care, particularly relevant due to COVID-19. However, few studies [105] analyze the health care industry, which deserves further in-depth analyses. According to Bhatnagar et al. [106], the alignment of sustainability with economic goals in the sustainable business model innovation process is fundamental to value creation for stakeholders, but further studies on the different phases of sustainable business model innovation would be appropriate. Another research trend is related to circular business models [107], which have a main quality of pursuing three different aims: reducing waste, increasing environmental performance, and increasing economic benefits [108]. Specifically, the business model assumes a new identity due to strategy implementations that intend to pursue environmental and socio-economic performance [109], generating a shift requiring a full transformation of the business processes that circular business models can realize. However, only some industries have received the right attention [110–113]. Other studies investigated the role of smart technologies in

the creation of sustainable business models [114] and analyzed how sustainable business models are the result of smart technology's impact on the change management process [2].

Regarding the third group, "green supply chain, logistics and digital management", several research trends emerge. A first trend can be attributed to the resilience and agility of supply chains viable thanks to the enabler role of digital technologies, and strongly emerged due to COVID-19. Previously, authors as Eltantawy [115] have shown that this resilience is a crucial capability to offer sustainable supply chains the opportunity to support their economic, environmental, and social performance. Frederico [116] analyzes how disruptive technologies are essential to respond to the COVID-19 pandemic, by underlining that supply chain 4.0, identified as a transformational strategic development, is effective to trigger resilience. From the analysis of results, it emerges that there is a need to eliminate influential barriers in digital supply chain development [117] to transform the traditional supply chain into a sustainable digital supply chain, considering sustainability an integral part of transformation process to improve decision-making activities based on knowledge-based systems and to make possible coordination and collaboration across stakeholders [118]. This transformation of the supply chain is necessary to reach a high level of sustainability. Concerning this point, it would be appropriate to pay attention to supply chain finance by adopting both a finance-oriented perspective and a supply-chainoriented perspective. "By constituting a critical part of supply chain management that connects buyers, sellers, and financial institutions" [119], it can contribute to the creation of a sustainable supply chain for firms' success.

Environmental sustainability is an imperative entailing a change in the value creation process [120] and the adjustment of supply chains to embrace circular economy principles [117], but it would be appropriate to deepen the strategies promoting adoption of CE principles related to digital technologies in the supply chain. A second research trend concerns the relevance of supply chain transparency and viability [121,122], in that it does not only regard direct suppliers and customers but involves multiple level of supply chains. Even if supply chain transparency (SCT) is widely examined in literature, as noted also by Ebinger and Omondi [123], it is necessary to continue exploring the field in relation to the integration of the different digital approaches to support the achievement of transparency. Furthermore, few studies explore the role of suppliers and how their different positions impact supply chain transparency. The main studies analyze the agrifood industry [120,124–127], fashion online retail [128], the manufacturing sector [129,130], healthcare [131], and the fuel and electronics industry [132–134]. The third research trend can be recognized in the influence of the choice of logistics service providers on the sustainable performance of supply chains, and, although there are some studies that explore the sustainable service quality as an attribute to evaluate logistics providers [135], there is still a fragmented and incomplete framework.

Regarding the fourth group, "sustainable financial and economic performance, value creation, and digitization", it mainly highlights the correlation between economic and environmental performance, which depends on firm external variables (regulations, for example) and internal variables affected by management and their choices. The decision-making perspective takes on a fundamental role, since it is able to influence the relationship, which is described as a U-shaped curve [136]. Even if many studies analyze the relationship above, showing the positive influence of sustainable strategy on financial performance [137], understanding the mechanisms at the foundation of this impact remains a question that deserves further investigation, because there is still a lack of consensus on the specific kinds of sustainability [138].

Regarding the fifth group, "sustainable entrepreneurship and digital technologies", the following trends can be identified. One is the linkage between entrepreneurial attitudes for twinning between digitalization and sustainability in regard to the attitudes that drive the actions; according to Ferreira et al. [139], there is a mutual relationship that links digital and environmental transformation with entrepreneurship, since the aforementioned variables are not mutually exclusive, but can coexist. Within this line of inquiry,

a main distinction between environmental entrepreneurship and a green entrepreneurial orientation [140] is observed, highlighting an inequality in the extent of investigations on the green entrepreneurial orientation. In fact, few studies focus on this issue [140,141]. Previously, authors as Lumpkin and Dess [142] introduced the entrepreneurial orientation as a construct that is differentiated from entrepreneurial intensity (EI). If the first is defined as a set of processes, practices, and decision-making activities for the new firms' creation, EI focuses on "the degree of entrepreneurship, the level of commitment and focus in leading a new entry" [143]. However, this relationship is little developed so far. Therefore, a second research trend could be related to the linkage between entrepreneurial intensity and sustainability [144] and could analyze how technological advances play a crucial role in the relationship, of which the aim is to intensify the firm's capability to learn innovative entrepreneurial ways to guarantee responsible performance. Another research trend concerns the analysis of digital technology start-ups to comprehend the extent of their contribution to sustainable transformation [145] and to achieve SDGs. These start-ups tend to direct their attention to economic success, moving sustainable goals to the background. For this reason, further in-depth analyses are necessary to explore how the start-ups perceive the sustainability concept. Biancone et al. [146] outline that, in the global scenario, technology's elements become fundamentals for entrepreneurs if they intend to address the challenge of the global markets. The COVID-19 emergency dictated new conditions of uncertainty, manageable thanks to the transformation of traditional entrepreneurship into digital entrepreneurship to achieve sustainable development. Therefore, firms should start this transformational process. The COVID-19 pandemic was a turning point in the ways of doing business and laid the foundations for searching new ways to interpret entrepreneurship from a digital perspective, pointing to sustainable recovery through a shared-resource model. Looking at the results, to enrich the existing literature with more current scenario analyses seems appropriate. This could make it possible to understand how entrepreneurial resilience can facilitate the overcoming of these uncertainties.

Regarding the sixth group, "circular economy and digitization", this includes the main topic about the circular economy issue and investigates barriers and drivers supporting the transition to a CE. Kayikci et al. [147] observe that governmental support and administrative burden are fundamental to promoting circular economy; furthermore, they note the lack of effective execution of environmental regulations is a barrier to this transition. Other barriers are related to ineffective CE framework adoption, and, finally, to the lack of integration and collaboration between supply chain participants. Abdul-Hamid et al. [148] define a set of industry 4.0 drivers in the circular economy for the palm oil industry, among which business model innovation, laws and policy, workforce empowerment, government support, and waste management emerge. Agyemang et al. [149], instead, observe that profitability/market share/benefit; reduction in cost; resource efficiency; environmental safety and risk management of health and safety issues; and product quality represent some of the main drivers identified as internal drivers capable to facilitate the CE transition, while international competition and pressure, stakeholder pressure, and support from parent companies represent the external drivers. From the analysis of results, it emerges that it is necessary to produce more research to develop a holistic view of circular manufacturing based on the structuring of data and information able to sustain the decision-making processes of manufacturers that intend to achieve great circular performance [150]. The increasing attention to the CE domain and digital technologies is linked to the impact that they are able to generate on practices for logistics and supply chain management [126,151]. In fact, if manufacturing organizations aspire to be competitive on an international scale, they should direct their view to Industry 4.0 and circular economy practices [151]. Therefore, firms should search for the best way to adopt these technology tools to build an integrated framework in which they develop an operative action plan for sustainable activity implementation [152]. The CE transition also deserves more investigation, especially for emerging economies, since there are still few studies which focus on countries like Pakistan [149]. Furthermore, if, on one hand, the Social Organizational Life Cycle Assessment (SO-LCA) methodology is used to correlate social impact categories and subcategories with organization-specific social metrics [153], on the other hand, further research should deepen how digitalization of production processes takes on an essential role in knowing social performance. The link between firms' organizational capabilities and their digital and circular strategies is another trend emerging from our results. Kristoffersen et al. [154,155] highlight that there is a need to develop organizational capabilities making it possible to utilize digital technologies like business analytics to increase their ability to pursue circular strategies.

# 6. Conclusions, Future Research Directions, Implications and Limitations

# 6.1. Conclusions

The relationship between sustainability and digitalization, explored through an SLR methodology that used a sample of scholarly works published over the period 2019–2023, made it possible to summarize the following results.

In response to RQ1 and RQ2, seven research groups can be delineated:

In each group, the contributions increased over recent years, confirming the relevance of digitization for sustainability from a decisional perspective. However, the results indicate that the literature is focused on only some aspects of the relationship between sustainability and digitalization, and different gaps deserve to be filled. Specifically, the SLR results show that groups A1., A3., and A4., due to the number of contributions, prevail over others, even if the sustainable financial performance and value creation cluster contains only 37 articles focused on sustainability, which is interpreted as an imperative for firms that intend to remain competitive in a sustainable future possible through digital technology adoption. Therefore, more studies in the field are required [156]. Instead, less attention is paid to the link between sustainable entrepreneurship and digitalization and relative change to generate the sustainability-oriented entrepreneurship.

In agreement with Denicolai et al. [157], digital transformation is taking on a more and more relevant role in the current era, becoming a central driver for innovation and business renewal. In fact, digital technologies are able to support sustainable business model creation by promoting an increase in productivity and a decrease in costs and emissions. From the SLR conducted, possible future research directions emerge. To answer RQ3 ("What are the possible future directions of research?") in the next lines, possible future directions are detailed for each research cluster, after a brief summary of the results.

A1. This is one of the richest strands of research. From analysis of the literature, it emerges that the concept of sustainable development requires changes in management to improve its approach to sustainability and to promote the observance of guidelines established from Agenda 2030. Hence, future research could:

- Define a set of green indicators that can facilitate and improve the life cycle of sustainability by requiring a holistic decision-making process.
- 2. Investigate the use of technology in sustainable social businesses by analyzing the interaction between different dimensions of technology adoption and social firms, since, to the best of the authors' knowledge, and in agreement with Soni et al. [158], there are no articles regarding this relationship.

A2. This cluster focuses on business model innovation to promote sustainability. The analysis brought to light that contributions on the sustainable business model are considerably increasing, mainly investigating the issue through the exploration of the relevant role that it assumes to satisfy economic, environmental, and social aims. Analysis of the literature outlines the following proposals for future research:

- 1. Scouting the adoption of a sustainable business model that entails relevant challenges that the whole organization must deal with. For this reason, it would be useful to analyze how the transition from a traditional business model to a sustainable business model [159] in different industries occurs and to measure its impact on decision-making processes;
- 2. Defining shared best practices for sustainable business model adoption;

3. Analyzing the consequences of management of the COVID-19 pandemic emergency on sustainable business models' role.

A3. This cluster refers to changes in supply chain from the sustainability perspective. In particular, this group covers all the articles that study the interconnection between sustainability, supply chains, and digitalization, starting from the consideration that logistics is defined as "execution of supply chain management activities" [160]. The analysis of the literature suggests the following future research:

- 1. In-depth analysis of critical circular supply chain management and related indicators, in order to shed light on the best performance in sustainable circular supply chains that the measurements make it possible to identify [130].
- 2. Investigation of the impact of digital transformation on sustainable supply chains by analyzing the collaboration between different actors as a critical success factor for sustainable supply chain management [161].

A4. This cluster gathers publications focused on financial performance improvement. From the literature analysis, the following, not exhaustive, proposals for future research can be outlined:

- 1. Shed light on the relationship between financial performance and sustainable digital performance;
- 2. Improve the evaluation of organizations through financial and sustainable indicators to compare profit and non-profit organizations;
- 3. Categorize suitable financial measures and analyze them to verify their reliability for sustainable value creation;
- 4. Focus on moderating factors that impact the relationship between financial and sustainable performance.

A5. This group includes studies that analyze the link between sustainable entrepreneurship and digitalization, focusing on the necessary changes to generate sustainabilityoriented entrepreneurship, and they reveal the new concept of the sustainable entrepreneurial model [162]. The literature analysis points out that researchers should investigate barriers and enablers to promote the sustainable entrepreneurial model and evaluate the perceived financial risks for this model adoption.

A6. This group includes articles on circular economy. It mainly investigates digital barriers and drivers for the transition to a CE. In agreement with Burmaoglu et al. [163], further exploring in this direction is necessary to understand how the CE perspective can be embedded in digital skills and aid firms to become sustainable.

## 6.2. Implications

The present study has several implications for scholars and practitioners. First, this SLR offers a comprehensive overview and a critical reflection on the sustainability approach and tries to define how digitalization influences every single business dimension. Furthermore, the decision-making perspective adopted highlights the role of digitalization in the capture and value creation process [110]. In particular, the analysis is able to approve the proposition under which digital technologies are an efficient tool to create and preserve a sustainable firm, and more generally, a sustainable society for a long time. Based on the decision-making perspective, this study offers an integrated analysis investigating why and how being smart also means being sustainable [60], by contributing to identifying drivers, barriers, and indicators to building a sustainable path directed toward sustainable development. Moreover, this study indicates that new business model adoption can help firms to improve their sustainable and financial performance and help them by facilitating their transition to the circular economy, leaving behind the traditional business model. Finally, last but not least, the research tries to explain the relevance of digital technology implementation able to modify the management of supply chain structures [6].

#### 6.3. Limitations

This research suffers from some limitations. First of all, the use of a single database: even though Scopus is one of the most relevant databases for scholarly literature, other important databases, such as Google Scholar and Web of Science, could be queried for future enlargement of the study. Secondly, the manual clustering applied implies a degree of subjectivity that could impact the results. Thirdly, the relatively short period considered might influence the outcome of the analysis. Finally, the analysis could also be associated with an empirical methodology to consider the influence of digitization on the sustainabilityoriented business model and sustainable decision-making processes.

To conclude, although this study intends to increase attention to the relationship between digitalization and sustainability, further analyses are necessary to fully understand the role played by digitalization and its capability to respond to social challenges in terms of sustainable development. Regarding this point, and considering the importance of SDGs, it could be useful to examine how the decision-making perspective can take on a fundamental role to increase and spread sustainable development.

Author Contributions: Conceptualization, T.R.; methodology, T.R.; formal analysis, C.M.; data curation, C.M.; writing—original draft preparation, T.R.; writing—review and editing, C.M. All authors have read and agreed to the published version of the manuscript.

Funding: This research received no external funding.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Conflicts of Interest: The authors declare no conflict of interest.

# References

- 1. Li, N.; Wang, X.; Zhang, S. Effects of Digitization on Enterprise Growth Performance: Mediating Role of Strategic Change and Moderating Role of Dynamic Capability. *Manag. Decis. Econ.* **2022**. [CrossRef]
- 2. Fiorentino, R.; Grimaldi, F.; Lamboglia, R.; Merendino, A. How Smart Technologies Can Support Sustainable Business Models: Insights from an Air Navigation Service Provider. *Manag. Decis.* **2020**, *58*, 1715–1736. [CrossRef]
- Merendino, A.; Dibb, S.; Meadows, M.; Quinn, L.; Wilson, D.; Simkin, L.; Canhoto, A. Big Data, Big Decisions: The Impact of Big Data on Board Level Decision-Making. J. Bus. Res. 2018, 93, 67–78. [CrossRef]
- 4. Kristoffersen, E.; Blomsma, F.; Mikalef, P.; Li, J. The Smart Circular Economy: A Digital-Enabled Circular Strategies Framework for Manufacturing Companies. J. Bus. Res. 2020, 120, 241–261. [CrossRef]
- 5. Antikainen, M.; Uusitalo, T.; Kivikytö-Reponen, P. Digitalisation as an Enabler of Circular Economy. *Procedia CIRP* 2018, 73, 45–49. [CrossRef]
- Khan, I.S.; Ahmad, M.O.; Majava, J. Industry 4.0 and Sustainable Development: A Systematic Mapping of Triple Bottom Line, Circular Economy and Sustainable Business Models Perspectives. J. Clean. Prod. 2021, 297, 126655. [CrossRef]
- Domenech, T.; Fokeer, S. Why Innovative Manufacturing and Circularity Are Key for a Resilient Manufacturing Industry Post-COVID-19 | UNIDO. Available online: https://www.unido.org/news/why-innovative-manufacturing-and-circularity-arekey-resilient-manufacturing-industry-post-COVID-19 (accessed on 27 November 2022).
- Shilo, S.; Rossman, H.; Segal, E. Axes of a Revolution: Challenges and Promises of Big Data in Healthcare. *Nat. Med.* 2020, 26, 29–38. [CrossRef]
- 9. Heaven, D. Bitcoin for the Biological Literature. Nature 2019, 566, 141–142. [CrossRef]
- 10. Perkel, J.M. The Internet of Things Comes to the Lab. Nature 2017, 542, 125–126. [CrossRef]
- Magni, D.; Piccolo, R.; Scuotto, V.; Papa, A.; Giudice, M.D. Immersive Technologies and Smart Cities in ASEAN: Discovering the Future Innovation Challenges. Available online: https://www.igi-global.com/chapter/immersive-technologies-and-smart-cities-inasean/www.igi-global.com/chapter/immersive-technologies-and-smart-cities-in-asean/275906 (accessed on 27 November 2022).
- 12. Sambamurthy, V.; Bharadwaj, A.; Grover, V. Shaping Agility through Digital Options: Reconceptualizing the Role of Information Technology in Contemporary Firms. *MIS Q.* **2003**, *27*, 237–263. [CrossRef]
- Merrill, R.K.; Schillebeeckx, S.J.D.; Blakstad, S. Sustainable digital finance in Asia: Creating environmental impact through bank transformation. SDFA, DBS, UN Environment 2019. Available online: https://www.dbs.com/iwov-resources/images/ sustainability/insights/Sustainable%20Digital%20Finance%20in%20Asia\_FINAL\_22.pdf?pid=sg-group-pweb-sustainabilitypdf-Sustainable%20Digital%20Finance%20in%20Asia\_FINAL\_22 (accessed on 27 November 2022).

- 14. Luthra, S.; Mangla, S.K. Evaluating Challenges to Industry 4.0 Initiatives for Supply Chain Sustainability in Emerging Economies. *Process Saf. Environ. Prot.* 2018, 117, 168–179. [CrossRef]
- 15. Di Vaio, A.; Syriopoulos, T.; Alvino, F.; Palladino, R. "Integrated Thinking and Reporting" towards Sustainable Business Models: A Concise Bibliometric Analysis. *Meditari Account. Res.* **2020**, *29*, 691–719. [CrossRef]
- 16. Paoloni, P.; Modaffari, G.; Ricci, F.; Della Corte, G. Intellectual Capital between Measurement and Reporting: A Structured Literature Review. *J. Intellect. Cap.* **2022**. *ahead-of-print*. [CrossRef]
- 17. Secundo, G.; Ndou, V.; Vecchio, P.D.; De Pascale, G. Sustainable Development, Intellectual Capital and Technology Policies: A Structured Literature Review and Future Research Agenda. *Technol. Forecast. Soc. Change* **2020**, *153*, 119917. [CrossRef]
- Dada, O. (Lola) A Model of Entrepreneurial Autonomy in Franchised Outlets: A Systematic Review of the Empirical Evidence. Int. J. Manag. Rev. 2018, 20, 206–226. [CrossRef]
- Massaro, M.; Dumay, J.; Garlatti, A. Public Sector Knowledge Management: A Structured Literature Review. J. Knowl. Manag. 2015, 19, 530–558. [CrossRef]
- Massaro, M.; Dumay, J.; Guthrie, J. On the Shoulders of Giants: Undertaking a Structured Literature Review in Accounting. Account. Audit. Account. J. 2016, 29, 767–801. [CrossRef]
- Secinaro, S.; Calandra, D.; Secinaro, A.; Muthurangu, V.; Biancone, P. The Role of Artificial Intelligence in Healthcare: A Structured Literature Review. BMC Med. Inform. Decis. Mak. 2021, 21, 125. [CrossRef]
- 22. De Felice, F.; Petrillo, A. An Interdisciplinary Framework to Define Strategies for Digitalization and Sustainability: Proposal of a 'Digicircular' Model. *IET Collab. Intell. Manuf.* **2021**, *3*, 75–84. [CrossRef]
- 23. United Nations Brundtland Commission Our Common Future: Report of the World Commission on Environment and Development; United Nations Brundtland Commission: New York, NY, USA, 1987.
- 24. Soni, V.; Raizada, P.; Singh, P.; Cuong, H.N.; Rangabhashiyam, S.; Saini, A.; Saini, R.V.; Le, Q.V.; Nadda, A.K.; Le, T.-T.; et al. Sustainable and Green Trends in Using Plant Extracts for the Synthesis of Biogenic Metal Nanoparticles toward Environmental and Pharmaceutical Advances: A Review. *Environ. Res.* **2021**, 202, 111622. [CrossRef] [PubMed]
- 25. EU-OSHA The Circular Economy and Safety and Health: The Role of Digitalisation in the Circular Economy and Implications for Occupational Safety and Health until 2040 | Safety and Health at Work EU-OSHA; 2021. Available online: https://osha.europa.eu/en/publications/circular-economy-and-safety-and-health-role-digitalisation-circular-economy-and-implications-occupational-safety-and-health-until-2040 (accessed on 15 November 2021).
- 26. McKee, M.; Van Schalkwyk, M.C.I.; Stuckler, D. The Second Information Revolution: Digitalization Brings Opportunities and Concerns for Public Health. *Eur. J. Public Health* **2019**, *29*, 3–6. [CrossRef] [PubMed]
- Almaiah, M.A.; Alfaisal, R.; Salloum, S.A.; Al-Otaibi, S.; Al Sawafi, O.S.; Al-Maroof, R.S.; Lutfi, A.; Alrawad, M.; Mulhem, A.A.; Awad, A.B. Determinants Influencing the Continuous Intention to Use Digital Technologies in Higher Education. *Electronics* 2022, 11, 2827. [CrossRef]
- Akbari, M.; Hopkins, J.L. Digital Technologies as Enablers of Supply Chain Sustainability in an Emerging Economy. *Oper. Manag. Res.* 2022, 15, 689–710. [CrossRef]
- Morrone, C.; Attias, L.; Battisti, D.; Iacono, G. Italian Digital Transformation Team: The Relevance of the Digital Awareness. In Proceedings of the International Scientific Conference Economics of Digital Transformation Conference Proceeding, Rijeka, Opatija, Croatia, 25 June 2021.
- Capurro, R.; Fiorentino, R.; Garzella, S.; Giudici, A. Big Data Analytics in Innovation Processes: Which Forms of Dynamic Capabilities Should Be Developed and How to Embrace Digitization? *Eur. J. Innov. Manag.* 2021, 25, 273–294. [CrossRef]
- 31. Ghobakhloo, M. Industry 4.0, Digitization, and Opportunities for Sustainability. J. Clean. Prod. 2020, 252, 119869. [CrossRef]
- 32. Vaidya, S.; Ambad, P.; Bhosle, S. Industry 4.0—A Glimpse. Procedia Manuf. 2018, 20, 233–238. [CrossRef]
- 33. *Handbook of Digital Innovation;* Nambisan, S.; Lyytinen, K.; Yoo, Y. (Eds.) Edward Elgar Publishing: Cheltenham, UK; Northampton, MA, USA, 2020; ISBN 978-1-78811-997-9.
- 34. Bag, S.; Gupta, S.; Foropon, C. Examining the Role of Dynamic Remanufacturing Capability on Supply Chain Resilience in Circular Economy. *Manag. Decis.* 2018, 57, 863–885. [CrossRef]
- Bag, S.; Sahu, A.K.; Kilbourn, P.; Pisa, N.; Dhamija, P.; Sahu, A.K. Modeling Barriers of Digital Manufacturing in a Circular Economy for Enhancing Sustainability. *Int. J. Product. Perform. Manag.* 2021, 71, 833–869. [CrossRef]
- Korhonen, J.; Nuur, C.; Feldmann, A.; Birkie, S.E. Circular Economy as an Essentially Contested Concept. J. Clean. Prod. 2018, 175, 544–552. [CrossRef]
- Franzò, S.; Urbinati, A.; Chiaroni, D.; Chiesa, V. Unravelling the Design Process of Business Models from Linear to Circular: An Empirical Investigation. *Bus. Strategy Environ.* 2021, 30, 2758–2772. [CrossRef]
- Pieroni, M.P.P.; McAloone, T.C.; Pigosso, D.C.A. Business Model Innovation for Circular Economy and Sustainability: A Review of Approaches. J. Clean. Prod. 2019, 215, 198–216. [CrossRef]
- Geissdoerfer, M.; Pieroni, M.P.P.; Pigosso, D.C.A.; Soufani, K. Circular Business Models: A Review. J. Clean. Prod. 2020, 277, 123741. [CrossRef]
- 40. Saura, J.R.; Ribeiro-Soriano, D.; Palacios-Marqués, D. Adopting Digital Reservation Systems to Enable Circular Economy in Entrepreneurship. *Manag. Decis.* 2022. *ahead-of-print*. [CrossRef]

- Nascimento, D.L.M.; Alencastro, V.; Quelhas, O.L.G.; Caiado, R.G.G.; Garza-Reyes, J.A.; Rocha-Lona, L.; Tortorella, G. Exploring Industry 4.0 Technologies to Enable Circular Economy Practices in a Manufacturing Context: A Business Model Proposal. J. Manuf. Technol. Manag. 2018, 30, 607–627. [CrossRef]
- 42. Rosa, P.; Sassanelli, C.; Urbinati, A.; Chiaroni, D.; Terzi, S. Assessing Relations between Circular Economy and Industry 4.0: A Systematic Literature Review. *Int. J. Prod. Res.* 2020, *58*, 1662–1687. [CrossRef]
- 43. Stock, T.; Seliger, G. Opportunities of Sustainable Manufacturing in Industry 4.0. Procedia CIRP 2016, 40, 536–541. [CrossRef]
- 44. Massaro, M.; Secinaro, S.; Dal Mas, F.; Brescia, V.; Calandra, D. Industry 4.0 and Circular Economy: An Exploratory Analysis of Academic and Practitioners' Perspectives. *Bus. Strategy Environ.* **2021**, *30*, 1213–1231. [CrossRef]
- 45. Agrawal, R.; Wankhede, V.A.; Kumar, A.; Upadhyay, A.; Garza-Reyes, J.A. Nexus of Circular Economy and Sustainable Business Performance in the Era of Digitalization. *Int. J. Product. Perform. Manag.* **2021**, *71*, 748–774. [CrossRef]
- 46. Liao, Y.; Deschamps, F.; Loures, E.d.F.R.; Ramos, L.F.P. Past, Present and Future of Industry 4.0—A Systematic Literature Review and Research Agenda Proposal. *Int. J. Prod. Res.* 2017, *55*, 3609–3629. [CrossRef]
- 47. Di Maria, E.; De Marchi, V.; Galeazzo, A. Industry 4.0 Technologies and Circular Economy: The Mediating Role of Supply Chain Integration. *Bus. Strategy Environ.* 2022, *31*, 619–632. [CrossRef]
- Ghobakhloo, M. The Future of Manufacturing Industry: A Strategic Roadmap toward Industry 4.0. J. Manuf. Technol. Manag. 2018, 29, 910–936. [CrossRef]
- Ghobakhloo, M.; Fathi, M. Corporate Survival in Industry 4.0 Era: The Enabling Role of Lean-Digitized Manufacturing. J. Manuf. Technol. Manag. 2019, 31, 1–30. [CrossRef]
- Santos, C.; Mehrsai, A.; Barros, A.C.; Araújo, M.; Ares, E. Towards Industry 4.0: An Overview of European Strategic Roadmaps. Procedia Manuf. 2017, 13, 972–979. [CrossRef]
- 51. Witjes, S.; Lozano, R. Towards a More Circular Economy: Proposing a Framework Linking Sustainable Public Procurement and Sustainable Business Models. *Resour. Conserv. Recycl.* **2016**, *112*, 37–44. [CrossRef]
- 52. Cagno, E.; Neri, A.; Negri, M.; Bassani, C.A.; Lampertico, T. The Role of Digital Technologies in Operationalizing the Circular Economy Transition: A Systematic Literature Review. *Appl. Sci.* **2021**, *11*, 3328. [CrossRef]
- Pagoropoulos, A.; Pigosso, D.C.A.; McAloone, T.C. The Emergent Role of Digital Technologies in the Circular Economy: A Review. *Procedia CIRP* 2017, 64, 19–24. [CrossRef]
- 54. Hopkins, J.L. An Investigation into Emerging Industry 4.0 Technologies as Drivers of Supply Chain Innovation in Australia. *Comput. Ind.* **2021**, *125*, 103323. [CrossRef]
- 55. Ricci, F.; Scafarto, V.; Ferri, S.; Tron, A. Value Relevance of Digitalization: The Moderating Role of Corporate Sustainability. An Empirical Study of Italian Listed Companies. *J. Clean. Prod.* **2020**, *276*, 123282. [CrossRef]
- 56. Chaouni Benabdellah, A.; Zekhnini, K.; Cherrafi, A. Sustainable and Resilience Improvement through the Design for Circular Digital Supply Chain. In *Proceedings of the Advances in Production Management Systems*; Artificial Intelligence for Sustainable and Resilient Production Systems; Dolgui, A., Bernard, A., Lemoine, D., von Cieminski, G., Romero, D., Eds.; Springer International Publishing: Cham, Switzerland, 2021; pp. 550–559.
- 57. Aranda-Usón, A.; Portillo-Tarragona, P.; Scarpellini, S.; Llena-Macarulla, F. The Progressive Adoption of a Circular Economy by Businesses for Cleaner Production: An Approach from a Regional Study in Spain. J. Clean. Prod. 2020, 247, 119648. [CrossRef]
- 58. Junge, A.L.; Straube, F. Sustainable Supply Chains—Digital Transformation Technologies' Impact on the Social and Environmental Dimension. *Procedia Manuf.* 2020, 43, 736–742. [CrossRef]
- 59. Nidumolu, R.; Prahalad, C.K.; Rangaswami, M.R. Why Sustainability Is Now the Key Driver of Innovation. *Harv. Bus. Rev.* 2009, *87*, 57–64.
- 60. del Vecchio, P.; Malandugno, C.; Passiante, G.; Sakka, G. Circular Economy Business Model for Smart Tourism: The Case of Ecobnb. *EMJB* **2022**, *17*, 88–104. [CrossRef]
- Mahroof, K.; Omar, A.; Kucukaltan, B. Sustainable Food Supply Chains: Overcoming Key Challenges through Digital Technologies. Int. J. Product. Perform. Manag. 2021, 71, 981–1003. [CrossRef]
- 62. Urbinati, A.; Bogers, M.; Chiesa, V.; Frattini, F. Creating and Capturing Value from Big Data: A Multiple-Case Study Analysis of Provider Companies. *Technovation* **2019**, *84–85*, 21–36. [CrossRef]
- 63. Garnsey, E.; Hang, C.C. Introduction to Special Issue of Technovation on Opportunity Recognition and Creation. *Technovation* **2015**, *39*–40, 1–3. [CrossRef]
- 64. Bouncken, R.B.; Kraus, S.; Roig-Tierno, N. Knowledge- and Innovation-Based Business Models for Future Growth: Digitalized Business Models and Portfolio Considerations. *Rev. Manag. Sci.* 2021, *15*, 1–14. [CrossRef]
- 65. Geissdoerfer, M.; Morioka, S.N.; de Carvalho, M.M.; Evans, S. Business Models and Supply Chains for the Circular Economy. J. *Clean. Prod.* 2018, 190, 712–721. [CrossRef]
- Lutfi, A.; Alkelani, S.N.; Al-Khasawneh, M.A.; Alshira'h, A.F.; Alshirah, M.H.; Almaiah, M.A.; Alrawad, M.; Alsyouf, A.; Saad, M.; Ibrahim, N. Influence of Digital Accounting System Usage on SMEs Performance: The Moderating Effect of COVID-19. *Sustainability* 2022, 14, 15048. [CrossRef]
- 67. George, G.; Merrill, R.K.; Schillebeeckx, S.J.D. Digital Sustainability and Entrepreneurship: How Digital Innovations Are Helping Tackle Climate Change and Sustainable Development. *Entrep. Theory Pract.* **2021**, *45*, 999–1027. [CrossRef]
- Wut, T.M.; Lee, D.; Ip, W.M.; Lee, S.W. Digital Sustainability in the Organization: Scale Development and Validation. *Sustainability* 2021, 13, 3530. [CrossRef]

- 69. Prakash, G.; Ambedkar, K. Digitalization of Manufacturing for Implanting Value, Configuring Circularity and Achieving Sustainability. J. Adv. Manag. Res. 2022. ahead-of-print. [CrossRef]
- Lobschat, L.; Mueller, B.; Eggers, F.; Brandimarte, L.; Diefenbach, S.; Kroschke, M.; Wirtz, J. Corporate Digital Responsibility. J. Bus. Res. 2021, 122, 875–888. [CrossRef]
- Morosan, C.; DeFranco, A. When Tradition Meets the New Technology: An Examination of the Antecedents of Attitudes and Intentions to Use Mobile Devices in Private Clubs. *Int. J. Hosp. Manag.* 2014, 42, 126–136. [CrossRef]
- 72. Ricondo, I.; Arrieta, J.A.; Aranguren, N. NPD Risk Management: Proposed Implementation to Increase New Product Success. In Proceedings of the 2006 IEEE International Technology Management Conference (ICE), Milan, Italy, 26–28 June 2006; pp. 1–8.
- 73. Simons, H.L. *The Dynamic Option Selection System: Analyzing Markets and Managing Risk*, 1st ed.; John Wiley & Sons Inc.: New York, NY, USA, 1999; ISBN 978-0-471-32051-7.
- 74. Müller, J.; Voigt, K.-I. Sustainable Industrial Value Creation in SMEs: A Comparison between Industry 4.0 and Made in China 2025. *Int. J. Precis. Eng. Manuf. Green Technol.* **2018**, *5*, 659–670. [CrossRef]
- 75. Mihalache, M.; Mihalache, O.R. A Decisional Framework of Offshoring: Integrating Insights from 25 Years of Research to Provide Direction for Future\*. *Decis. Sci.* 2016, 47, 1103–1149. [CrossRef]
- Tranfield, D.; Denyer, D.; Smart, P. Towards a Methodology for Developing Evidence-Informed Management Knowledge by Means of Systematic Review. Br. J. Manag. 2003, 14, 207–222. [CrossRef]
- 77. Wang, C.L.; Chugh, H. Entrepreneurial Learning: Past Research and Future Challenges. Int. J. Manag. Rev. 2014, 16, 24-61. [CrossRef]
- Kauppi, K.; Salmi, A.; You, W. Sourcing from Africa: A Systematic Review and a Research Agenda. Int. J. Manag. Rev. 2018, 20, 627–650. [CrossRef]
- Nofal, A.M.; Nicolaou, N.; Symeonidou, N.; Shane, S. Biology and Management: A Review, Critique, and Research Agenda. J. Manag. 2018, 44, 7–31. [CrossRef]
- Dumay, J.; Cai, L. A Review and Critique of Content Analysis as a Methodology for Inquiring into IC Disclosure. *J. Intellect. Cap.* 2014, 15, 264–290. [CrossRef]
- 81. de Moya-Anegón, F.; Chinchilla-Rodríguez, Z.; Vargas-Quesada, B.; Corera-Álvarez, E.; Muñoz-Fernández, F.J.; González-Molina, A.; Herrero-Solana, V. Coverage Analysis of Scopus: A Journal Metric Approach. *Scientometrics* **2007**, *73*, 53–78. [CrossRef]
- 82. Paoloni, P.; Demartini, P. Women in Management: Perspectives on a Decade of Research (2005–2015). *Palgrave Commun* **2016**, 2, 16094. [CrossRef]
- 83. Jesson, J.; Lacey, F. How to Do (or Not to Do) a Critical Literature Review. Pharm. Educ. 2006, 6, 139–148. [CrossRef]
- Rodrigues, A.; Rojo, C.A.; Bertolini, G.R.F. Formulação de estratégias competitivas por meio de análise de cenários na construção civil. *Production* 2012, 23, 269–282. [CrossRef]
- Zarte, M.; Pechmann, A.; Nunes, I.L. Knowledge Framework for Production Planning and Controlling Considering Sustainability Aspects in Smart Factories. J. Clean. Prod. 2022, 363, 132283. [CrossRef]
- Chen, S.; Eweje, G. Managing Tensions in Sustainable Development in Chinese and New Zealand Business Partnerships: Integrative Approaches. *Bus. Strategy Environ.* 2022, 31, 2568–2587. [CrossRef]
- Tura, N.; Ojanen, V.; Hanski, J. Innovations for Sustainability: Challenges of Utilising Sustainability-Related Knowledge. Int. J. Innov. Sustain. Dev. 2019, 13, 452–478. [CrossRef]
- de Villiers, C.; Kuruppu, S.; Dissanayake, D. A (New) Role for Business—Promoting the United Nations' Sustainable Development Goals through the Internet-of-Things and Blockchain Technology. J. Bus. Res. 2021, 131, 598–609. [CrossRef]
- Jiang, S.; Jakobsen, K.; Jaccheri, L.; Li, J. Blockchain and Sustainability: A Tertiary Study. In Proceedings of the 2021 IEEE/ACM International Workshop on Body of Knowledge for Software Sustainability (BoKSS), Madrid, Spain, 1–2 June 2021; pp. 7–8.
- Parmentola, A.; Petrillo, A.; Tutore, I.; Felice, F.D. Is Blockchain Able to Enhance Environmental Sustainability? A Systematic Review and Research Agenda from the Perspective of Sustainable Development Goals (SDGs). *Bus. Strategy Environ.* 2022, 31, 194–217. [CrossRef]
- 91. Varriale, V.; Cammarano, A.; Michelino, F.; Caputo, M. The Unknown Potential of Blockchain for Sustainable Supply Chains. *Sustainability* **2020**, *12*, 9400. [CrossRef]
- Adams, R.; Kewell, B.; Parry, G. Blockchain for Good? Digital Ledger Technology and Sustainable Development Goals. In Handbook of Sustainability and Social Science Research; Leal Filho, W., Marans, R.W., Callewaert, J., Eds.; World Sustainability Series; Springer International Publishing: Cham, Switzerland, 2018; pp. 127–140, ISBN 978-3-319-67122-2.
- Tomlinson, B.; Boberg, J.; Cranefield, J.; Johnstone, D.; Luczak-Roesch, M.; Patterson, D.J.; Kapoor, S. Analyzing the Sustainability of 28 'Blockchain for Good' Projects via Affordances and Constraints. *Inf. Technol. Dev.* 2021, 27, 439–469. [CrossRef]
- 94. Bai, C.; Sarkis, J. A Supply Chain Transparency and Sustainability Technology Appraisal Model for Blockchain Technology. *Int. J. Prod. Res.* **2020**, *58*, 2142–2162. [CrossRef]
- 95. Estevez, E.; Cenci, K.; Fillottrani, P.; Janowski, T. Review of International Standards and Policy Guidelines for Smart Sustainable Cities. In Smart Cities and Smart Governance: Towards the 22nd Century Sustainable City; Estevez, E., Pardo, T.A., Scholl, H.J., Eds.; Public Administration and Information Technology; Springer International Publishing: Cham, Switzerland, 2021; pp. 69–99, ISBN 978-3-030-61033-3.
- 96. Ligorio, L.; Venturelli, A.; Caputo, F. Tracing the Boundaries between Sustainable Cities and Cities for Sustainable Development. An LDA Analysis of Management Studies. *Technol. Forecast. Soc. Change* **2022**, *176*, 121447. [CrossRef]

- Guandalini, I. Sustainability through Digital Transformation: A Systematic Literature Review for Research Guidance. J. Bus. Res. 2022, 148, 456–471. [CrossRef]
- Bag, S.; Pretorius, J.H.C.; Gupta, S.; Dwivedi, Y.K. Role of Institutional Pressures and Resources in the Adoption of Big Data Analytics Powered Artificial Intelligence, Sustainable Manufacturing Practices and Circular Economy Capabilities. *Technol. Forecast. Soc. Change* 2021, 163, 120420. [CrossRef]
- 99. Jamwal, A.; Agrawal, R.; Sharma, M. Deep Learning for Manufacturing Sustainability: Models, Applications in Industry 4.0 and Implications. *Int. J. Inf. Manag. Data Insights* 2022, 2, 100107. [CrossRef]
- Kumar, V.; Vrat, P.; Shankar, R. Factors Influencing the Implementation of Industry 4.0 for Sustainability in Manufacturing. *Glob. J. Flex. Syst. Manag.* 2022, 23, 453–478. [CrossRef]
- Verma, P.; Kumar, V.; Daim, T.; Sharma, N.K.; Mittal, A. Identifying and Prioritizing Impediments of Industry 4.0 to Sustainable Digital Manufacturing: A Mixed Method Approach. J. Clean. Prod. 2022, 356, 131639. [CrossRef]
- 102. Fehrer, J.A.; Wieland, H. A Systemic Logic for Circular Business Models. J. Bus. Res. 2021, 125, 609–620. [CrossRef]
- Nardi, M.L. Ethics and Managerial Mindset in Politics. In Computational Thinking for Problem Solving and Managerial Mindset Training; IGI Global: Hershey, Pennsylvania, USA, 2021; pp. 252–265, ISBN 978-1-79987-128-6.
- 104. Fürstenau, D.; Klein, S.; Vogel, A.; Auschra, C. Multi-Sided Platform and Data-Driven Care Research. *Electron Mark.* 2021, 31, 811–828. [CrossRef]
- Oderanti, F.; Li, F.; Cubric, M.; Shi, X. Business Models for Sustainable Commercialisation of Digital Healthcare (EHealth) Innovations for an Increasingly Ageing Population (A New Business Model for EHealth). *Technol. Forecast. Soc. Change* 2021, 171, 120969. [CrossRef]
- Bhatnagar, R.; Keskin, D.; Kirkels, A.; Romme, A.G.L.; Huijben, J.C.C.M. Design Principles for Sustainability Assessments in the Business Model Innovation Process. J. Clean. Prod. 2022, 377, 134313. [CrossRef]
- 107. Frishammar, J.; Parida, V. Circular Business Model Transformation: A Roadmap for Incumbent Firms. *Calif. Manag. Rev.* 2019, 61, 5–29. [CrossRef]
- 108. Parida, V.; Sjödin, D.; Reim, W. Reviewing Literature on Digitalization, Business Model Innovation, and Sustainable Industry: Past Achievements and Future Promises. *Sustainability* **2019**, *11*, 391. [CrossRef]
- Dahmani, N.; Benhida, K.; Belhadi, A.; Kamble, S.; Elfezazi, S.; Jauhar, S.K. Smart Circular Product Design Strategies towards Eco-Effective Production Systems: A Lean Eco-Design Industry 4.0 Framework. J. Clean. Prod. 2021, 320, 128847. [CrossRef]
- Urbinati, A.; Franzò, S.; Chiaroni, D. Enablers and Barriers for Circular Business Models: An Empirical Analysis in the Italian Automotive Industry. *Sustain. Prod. Consum.* 2021, 27, 551–566. [CrossRef]
- 111. Huynh, P.H. Enabling Circular Business Models in the Fashion Industry: The Role of Digital Innovation. *Int. J. Product. Perform. Manag.* **2021**, *71*, 870–895. [CrossRef]
- 112. Urbinati, A.; Rosa, P.; Sassanelli, C.; Chiaroni, D.; Terzi, S. Circular Business Models in the European Manufacturing Industry: A Multiple Case Study Analysis. *J. Clean. Prod.* **2020**, 274, 122964. [CrossRef]
- 113. Upadhyay, A.; Akter, S.; Adams, L.; Kumar, V.; Varma, N. Investigating "Circular Business Models" in the Manufacturing and Service Sectors. *J. Manuf. Technol. Manag.* **2019**, *30*, 590–606. [CrossRef]
- Del Giudice, M.; Di Vaio, A.; Hassan, R.; Palladino, R. Digitalization and New Technologies for Sustainable Business Models at the Ship–Port Interface: A Bibliometric Analysis. *Marit. Policy Manag.* 2022, 49, 410–446. [CrossRef]
- Eltantawy, R. Towards Sustainable Supply Management: Requisite Governance and Resilience Capabilities. J. Strateg. Mark. 2016, 24, 118–130. [CrossRef]
- 116. Frederico, G.F. Towards a Supply Chain 4.0 on the Post-COVID-19 Pandemic: A Conceptual and Strategic Discussion for More Resilient Supply Chains. *Rajagiri Manag. J.* **2021**, *15*, 94–104. [CrossRef]
- Dwivedi, A.; Paul, S.K. A Framework for Digital Supply Chains in the Era of Circular Economy: Implications on Environmental Sustainability. *Bus. Strategy Environ.* 2022, 31, 1249–1274. [CrossRef]
- 118. Di Vaio, A.; Varriale, L. Blockchain Technology in Supply Chain Management for Sustainable Performance: Evidence from the Airport Industry. *Int. J. Inf. Manag.* 2020, *52*, 102014. [CrossRef]
- 119. Alsmadi, A.; Al-Gasaymeh, A.; Alrawashdeh, N.; Alhwamdeh, L. Financial Supply Chain Management: A Bibliometric Analysis for 2006–2022. *Uncertain Supply Chain Manag.* 2022, 10, 645–656. [CrossRef]
- 120. Kumar, S.; Raut, R.D.; Nayal, K.; Kraus, S.; Yadav, V.S.; Narkhede, B.E. To Identify Industry 4.0 and Circular Economy Adoption Barriers in the Agriculture Supply Chain by Using ISM-ANP. J. Clean. Prod. **2021**, 293, 126023. [CrossRef]
- 121. Schäfer, N. Making Transparency Transparent: A Systematic Literature Review to Define and Frame Supply Chain Transparency in the Context of Sustainability. *Manag. Rev. Q* 2022. [CrossRef]
- 122. Adhi Santharm, B.; Ramanathan, U. Supply Chain Transparency for Sustainability—An Intervention-Based Research Approach. Int. J. Oper. Prod. Manag. 2022, 42, 995–1021. [CrossRef]
- 123. Ebinger, F.; Omondi, B. Leveraging Digital Approaches for Transparency in Sustainable Supply Chains: A Conceptual Paper. *Sustainability* **2020**, *12*, 6129. [CrossRef]
- 124. Anastasiadis, F.; Manikas, I.; Apostolidou, I.; Wahbeh, S. The Role of Traceability in End-to-End Circular Agri-Food Supply Chains. *Ind. Mark. Manag.* 2022, 104, 196–211. [CrossRef]
- 125. Balezentis, T.; Zickiene, A.; Volkov, A.; Streimikiene, D.; Morkunas, M.; Dabkiene, V.; Ribasauskiene, E. Measures for the Viable Agri-Food Supply Chains: A Multi-Criteria Approach. J. Bus. Res. 2023, 155, 113417. [CrossRef]

- 126. Kharola, S.; Ram, M.; Kumar Mangla, S.; Goyal, N.; Nautiyal, O.P.; Pant, D.; Kazancoglu, Y. Exploring the Green Waste Management Problem in Food Supply Chains: A Circular Economy Context. J. Clean. Prod. **2022**, 351, 131355. [CrossRef]
- 127. Moysiadis, T.; Spanaki, K.; Kassahun, A.; Kläser, S.; Becker, N.; Alexiou, G.; Zotos, N.; Karali, I. AgriFood Supply Chain Traceability: Data Sharing in a Farm-to-Fork Case. *Benchmarking: Int. J.* **2022**. *ahead-of-print*. [CrossRef]
- 128. Pereira, A.M.; Moura, J.A.B.; Costa, E.D.B.; Vieira, T.; Landim, A.R.D.B.; Bazaki, E.; Wanick, V. Customer Models for Artificial Intelligence-Based Decision Support in Fashion Online Retail Supply Chains. *Decis. Support Syst.* **2022**, *158*, 113795. [CrossRef]
- Kamble, S.S.; Gunasekaran, A.; Parekh, H.; Mani, V.; Belhadi, A.; Sharma, R. Digital Twin for Sustainable Manufacturing Supply Chains: Current Trends, Future Perspectives, and an Implementation Framework. *Technol. Forecast. Soc. Change* 2022, 176, 121448. [CrossRef]
- 130. Saraji, M.K.; Streimikiene, D. Evaluating the Circular Supply Chain Adoption in Manufacturing Sectors: A Picture Fuzzy Approach. *Technol. Soc.* 2022, 70, 102050. [CrossRef]
- 131. Azadi, M.; Yousefi, S.; Farzipoor Saen, R.; Shabanpour, H.; Jabeen, F. Forecasting Sustainability of Healthcare Supply Chains Using Deep Learning and Network Data Envelopment Analysis. *J. Bus. Res.* **2023**, *154*, 113357. [CrossRef]
- Kunkel, S.; Matthess, M.; Xue, B.; Beier, G. Industry 4.0 in Sustainable Supply Chain Collaboration: Insights from an Interview Study with International Buying Firms and Chinese Suppliers in the Electronics Industry. *Resour. Conserv. Recycl.* 2022, 182, 106274. [CrossRef]
- 133. Matthess, M.; Kunkel, S.; Xue, B.; Beier, G. Supplier Sustainability Assessment in the Age of Industry 4.0—Insights from the Electronics Industry. *Clean. Logist. Supply Chain* **2022**, *4*, 100038. [CrossRef]
- Nocera Alves Junior, P.; Costa Melo, I.; de Moraes Santos, R.; da Rocha, F.V.; Caixeta-Filho, J.V. How Did COVID-19 Affect Green-Fuel Supply Chain?—A Performance Analysis of Brazilian Ethanol Sector. *Res. Transp. Econ.* 2022, 93, 101137. [CrossRef]
- Gupta, A.; Singh, R.K. Study of Sustainability Issues in an Indian Logistics Service Provider: SAP-LAP Approach. *Qual. Res. Organ. Manag. Int. J.* 2020, 16, 530–549. [CrossRef]
- 136. Boakye, D.J.; Tingbani, I.; Ahinful, G.S.; Nsor-Ambala, R. The Relationship between Environmental Management Performance and Financial Performance of Firms Listed in the Alternative Investment Market (AIM) in the UK. J. Clean. Prod. 2021, 278, 124034. [CrossRef]
- Kumar, S.; Dua, P. Environmental Management Practices and Financial Performance: Evidence from Large Listed Indian Enterprises. J. Environ. Plan. Manag. 2022, 65, 37–61. [CrossRef]
- Bartolacci, F.; Caputo, A.; Soverchia, M. Sustainability and Financial Performance of Small and Medium Sized Enterprises: A Bibliometric and Systematic Literature Review. Bus. Strategy Environ. 2020, 29, 1297–1309. [CrossRef]
- 139. Ferreira, J.J.; Fernandes, C.I.; Veiga, P.M.; Caputo, A. The Interactions of Entrepreneurial Attitudes, Abilities and Aspirations in the (Twin) Environmental and Digital Transitions? A Dynamic Panel Data Approach. *Technol. Soc.* **2022**, *71*, 102121. [CrossRef]
- 140. Ameer, F.; Khan, N.R. Green Entrepreneurial Orientation and Corporate Environmental Performance: A Systematic Literature Review. *Eur. Manag. J.* 2022. [CrossRef]
- 141. Pratono, A.H.; Darmasetiawan, N.K.; Yudiarso, A.; Jeong, B.G. Achieving Sustainable Competitive Advantage through Green Entrepreneurial Orientation and Market Orientation: The Role of Inter-Organizational Learning. *Bottom Line* 2019, 32, 2–15. [CrossRef]
- Lumpkin, G.T.; Dess, G.G. Clarifying the Entrepreneurial Orientation Construct and Linking It to Performance. *Acad. Manag. Rev.* 1996, 21, 135–172. [CrossRef]
- 143. Liao, J.; Murphy, P.J.; Welsch, H. Developing and Validating a Construct of Entrepreneurial Intensity. *New Engl. J. Entrep.* 2005, *8*, 31–38. [CrossRef]
- 144. Munyanyi, W.; Munongo, S.; Pooe, D. Spurring Entrepreneurial Intensity through Social Capital and Relationship Quality. S. Afr. J. Entrep. Small Bus. Manag. 2021, 13, 10. [CrossRef]
- Lammers, T.; Rashid, L.; Kratzer, J.; Voinov, A. An Analysis of the Sustainability Goals of Digital Technology Start-Ups in Berlin. *Technol. Forecast. Soc. Change* 2022, 185, 122096. [CrossRef]
- 146. Biancone, P.P.; Secinaro, S.; Iannaci, D.; Calandra, D. International Entrepreneurship and Technology: A Structured Literature Review. In *Empirical International Entrepreneurship: A Handbook of Methods, Approaches, and Applications*; Jafari-Sadeghi, V., Amoozad Mahdiraji, H., Dana, L.-P., Eds.; Contributions to Management Science; Springer International Publishing: Cham, Switzerland, 2021; pp. 137–159, ISBN 978-3-030-68972-8.
- 147. Kayikci, Y.; Kazancoglu, Y.; Lafci, C.; Gozacan, N. Exploring Barriers to Smart and Sustainable Circular Economy: The Case of an Automotive Eco-Cluster. J. Clean. Prod. 2021, 314, 127920. [CrossRef]
- 148. Abdul-Hamid, A.-Q.; Ali, M.H.; Osman, L.H.; Tseng, M.-L. The Drivers of Industry 4.0 in a Circular Economy: The Palm Oil Industry in Malaysia. J. Clean. Prod. 2021, 324, 129216. [CrossRef]
- 149. Agyemang, M.; Kusi-Sarpong, S.; Khan, S.A.; Mani, V.; Rehman, S.T.; Kusi-Sarpong, H. Drivers and Barriers to Circular Economy Implementation: An Explorative Study in Pakistan's Automobile Industry. *Manag. Decis.* **2019**, *57*, 971–994. [CrossRef]
- Acerbi, F.; Sassanelli, C.; Taisch, M. A Conceptual Data Model Promoting Data-Driven Circular Manufacturing. Oper. Manag. Res. 2022, 15, 838–857. [CrossRef]
- 151. Yadav, G.; Luthra, S.; Jakhar, S.K.; Mangla, S.K.; Rai, D.P. A Framework to Overcome Sustainable Supply Chain Challenges through Solution Measures of Industry 4.0 and Circular Economy: An Automotive Case. J. Clean. Prod. 2020, 254, 120112. [CrossRef]
- 152. Shang, C.; Saeidi, P.; Goh, C.F. Evaluation of Circular Supply Chains Barriers in the Era of Industry 4.0 Transition Using an Extended Decision-Making Approach. *J. Enterp. Inf. Manag.* **2022**, *35*, 1100–1128. [CrossRef]

- García-Muiña, F.; Medina-Salgado, M.S.; González-Sánchez, R.; Huertas-Valdivia, I.; Ferrari, A.M.; Settembre-Blundo, D. Industry 4.0-Based Dynamic Social Organizational Life Cycle Assessment to Target the Social Circular Economy in Manufacturing. J. Clean. Prod. 2021, 327, 129439. [CrossRef]
- 154. Kristoffersen, E.; Mikalef, P.; Blomsma, F.; Li, J. The Effects of Business Analytics Capability on Circular Economy Implementation, Resource Orchestration Capability, and Firm Performance. *Int. J. Prod. Econ.* **2021**, 239, 108205. [CrossRef]
- 155. Kristoffersen, E.; Mikalef, P.; Blomsma, F.; Li, J. Towards a Business Analytics Capability for the Circular Economy. *Technol. Forecast. Soc. Change* **2021**, *171*, 120957. [CrossRef]
- Hausberg, J.P.; Liere-Netheler, K.; Packmohr, S.; Pakura, S.; Vogelsang, K. Research Streams on Digital Transformation from a Holistic Business Perspective: A Systematic Literature Review and Citation Network Analysis. J. Bus. Econ. 2019, 89, 931–963. [CrossRef]
- 157. Denicolai, S.; Zucchella, A.; Magnani, G. Internationalization, Digitalization, and Sustainability: Are SMEs Ready? A Survey on Synergies and Substituting Effects among Growth Paths. *Technol. Forecast. Soc. Change* **2021**, *166*, 120650. [CrossRef]
- Soni, G.; Mangla, S.K.; Singh, P.; Dey, B.L.; Dora, M. Technological Interventions in Social Business: Mapping Current Research and Establishing Future Research Agenda. *Technol. Forecast. Soc. Change* 2021, 169, 120818. [CrossRef]
- 159. Nosratabadi, S.; Mosavi, A.; Shamshirband, S.; Kazimieras Zavadskas, E.; Rakotonirainy, A.; Chau, K.W. Sustainable Business Models: A Review. *Sustainability* **2019**, *11*, 1663. [CrossRef]
- 160. Lummus, R.R.; Krumwiede, D.W.; Vokurka, R.J. The Relationship of Logistics to Supply Chain Management: Developing a Common Industry Definition. *Ind. Manag. Data Syst.* 2001, 101, 426–432. [CrossRef]
- Tseng, M.-L.; Islam, M.S.; Karia, N.; Fauzi, F.A.; Afrin, S. A Literature Review on Green Supply Chain Management: Trends and Future Challenges. *Resour. Conserv. Recycl.* 2019, 141, 145–162. [CrossRef]
- 162. Xu, G.; Hou, G.; Zhang, J. Digital Sustainable Entrepreneurship: A Digital Capability Perspective through Digital Innovation Orientation for Social and Environmental Value Creation. *Sustainability* **2022**, *14*, 11222. [CrossRef]
- 163. Burmaoglu, S.; Ozdemir Gungor, D.; Kirbac, A.; Saritas, O. Future Research Avenues at the Nexus of Circular Economy and Digitalization. *Int. J. Product. Perform. Manag.* **2022**. *ahead-of-print*. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.